\input{ThisWeek.tex}
\newcommand{\duedate}{Thursday, September 11}
\input{../TemplateWeek/CheckHomeWork.tex}
\input{Exercises.tex}

%\booltrue{teacher}

\begin{enumerate}
\item You get a function $F(x)$ as the antiderivative.  $F(2)-F(1)$ should be about $-0.5022$.
\item $F(0)$ should be about $-20.6662$
\item You should have an unknown constant ($+C$ or something like that).  If you take $F(\pi/4)-F(0)$, the constants will cancel and you should get $3(\sqrt{2}-1$.
\item If you plug into a calculator, you should get about 2.0259
\item You should get $28/3$.
\end{enumerate}

\end{document}